Author’s response to reviews

Title: Progressive disseminated histoplasmosis with concomitant disseminated nontuberculous mycobacterial infection in a patient with AIDS from a nonendemic region (California).

Authors:

Julian Choi (julian.j.choi@gmail.com)
Kia Nikoomanesh (kianikoomanesh@gmail.com)
Jusleen Uppal (jusleenuppal@gmail.com)
Sharon Wang (Supra624@gmail.com)

Version: 1 Date: 24 Jan 2019

Author’s response to reviews:

To our reviewer, thank you for taking the time to look at our paper. We have looked over your analysis and made the changes you suggested. Please let us know if there is anything else that needs to be reviewed.

Thank you